
\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}

\def\udcs{517.5} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{41A05, 32A27} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}

\newtheorem*{problem*}{Problem}

%\theoremstyle{remark}
%\newtheorem{rem}{Remark}[section]
%\newtheorem{notation}{Notation}
%\renewcommand{\thenotation}{}
%\numberwithin{equation}{section}
%\newcommand{\interval}[1]{\mathinner{#1}}


\DeclareMathOperator*{\dm}{dim}
\DeclareMathOperator*{\res}{res}
\DeclareMathOperator*{\dif}{d}
\DeclareMathOperator*{\Res}{Res}
\DeclareMathOperator*{\LT}{LT}
\DeclareMathOperator*{\Lag}{Lag}





\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 16, стр. 144--144 (2019) \hspace{65mm}{\rm\small УДК \udcs}} %Volume, pages in Russian,
                                                                                    %filled by Editorial board
     \newline
         {\rm\small DOI~10.33048/semi.2019.16.xxx}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{1}




\bibitem{Alpay}\label{Alpay}
Alpay, D., Jorgensen, P., Lewkowicz, I. and Volok, D.,
\textit{A new realization of rational functions, with applications to linear combination interpolation, the Cuntz relations and kernel decompositions}, Complex Var. Elliptic Equ. -- 2016. -- no. 61(1). -- p. 42--54.

\bibitem{Yger}\label{Yger}
Alpay, D., Yger, A.,
\textit{About a Non-Standard Interpolation Problem}, Comput. Methods Funct. Theory 2019, 19, 97-115.

\bibitem{newAY}\label{newAY}
Alpay, D., Yger, A.,
\textit{Cauchy-Weil formula, Schur-Agler type classes, new Hardy spaces of the polydisk and interpolation problems}, Journal of Mathematical Analysis and Applications, vol. 504 (2), 2021, 125437.

\bibitem{BR}\label{BR}
De Boor, C., Ron, A.,
\textit{On multivariate polynomial interpolation}, Constructive Approximation, 1990, 6(3), p. 287 -- 302.

\bibitem{YV}\label{YV}
Vidras, A., Yger, A.,
\textit{Bergman-Weil expansion for holomorphic functions}, Mathematische Annalen, 2022, 382(6), p. 383 -- 419.


\bibitem{Tsikh}\label{Tsikh}
Tsikh, A.,
\textit{Multidimensional Residues and Their Applications (English summary)}, Translated from the 1988 Russian original by E. J. F. Primrose. Translations of Mathematical Monographs, 103. American Mathematical Society, Providence, RI, 1992. x+188 pp.

\bibitem{GH}\label{GH}
Griffiths, P.,  Harris, J., 
\textit{Principles of algebraic geometry}, Pure and Applied Mathematics. Wiley-Interscience [John Wiley and Sons], New York, 1978. xii+813 pp.


\bibitem{VDV}\label{VDV}
van der Waerden, B., 
\textit{Algebra. Vol. I.},  Based in part on lectures by E. Artin and E. Noether. Translated from the seventh German edition by Fred Blum and John R. Schulenberger. Springer-Verlag, New York, 1991. xiv+265 pp.

\bibitem{YA}\label{YA}
Aizenberg, I., Yuzhakov, A.,
\textit{Integral representations and residues in multidimensional complex analysis}, Translated from the Russian by H. H. McFaden. Translation edited by Lev J. Leifman. Translations of Mathematical Monographs, 58. American Mathematical Society, Providence, RI, 1983. x+283 pp.


\bibitem{LeTs}\label{LeTs}
Leinartas, E., Tsikh, A., 
\textit{On a Multidimentional Version of the Principal Theorem of Difference Equations with constant Coeffitients}, J. of Siberian Federal University. Math and Phys 15(1), 2022, 125 -- 132

\bibitem{Tajima}\label{Tajima}
Tajima S., Nabeshima K., 
\textit{Computing Grothendieck point residues via solving holonomic systems of first order partial differential equations}, ISSAC `21 Proceedings of the 2021 on International Symposium on Symbolic and Algebraic Computation, p. 361--368

\bibitem{Milnor}\label{Milnor}
Milnor, J., 
\textit{Singular points of complex hypersurfaces}, Annals of Mathematics Studies, no. 61, Princeton, New Jersey, 1968.


\bibitem{TY}\label{TY}
Tsikh, A., Yger,  A., 
\textit{Residue currents}, Complex analysis. J. Math. Sci. (N.Y.) 120 (2004), no. 6, p. 1916 -- 1971

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}


\thispagestyle{empty}

\title[Multidimensional Hermite interpolation]{Multidimensional Hermite interpolation
}
\author{{M.E. Durakov}}
\address{Matvey Durakov
\newline\hphantom{iii} Siberian Federal University,
\newline\hphantom{iii} pr. Svobodnyi, 79,
\newline\hphantom{iii} 660041, Krasnoyarsk, Russia}
\email{durakov\_m\_1997@mail.ru}


\author{{E.D. Leinartas}}
\address{Evgeniy Leinartas
\newline\hphantom{iii} Siberian Federal University,
\newline\hphantom{iii} pr. Svobodnyi, 79,
\newline\hphantom{iii} 660041, Krasnoyarsk, Russia}
\email{elyaynartas@sfu-kras.ru}


\author{{A.K. Tsikh}}
\address{August Tsikh
\newline\hphantom{iii} Siberian Federal University,
\newline\hphantom{iii} pr. Svobodnyi, 79,
\newline\hphantom{iii} 660041, Krasnoyarsk, Russia}
\email{atsikh@sfu-kras.ru}



\thanks{\sc Durakov, M.E., Leinartas, E.D., Tsikh, A.K.
Multidimensional Hermite interpolation}
\thanks{\copyright \ 2022 Durakov M.E., Leinartas E.D., Tsikh A.K.}
\thanks{\rm The investigation was supported by the Russian Science Foundation (grant No. 20-11-20117)}
\thanks{\it Received  January, 1, 2015, published  March, 1,  2015.}%


\semrtop \vspace{1cm}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } The Hermite interpolation formulas are based on the interpretation of interpolation nodes as roots of suitable polynomials. Therefore, such formulas belong to the class of algebraic interpolations. The article considers a multidimensional variant of Hermite interpolation, presents a class of algebraic systems of equations for which the Hermite interpolation polynomial is represented by an explicit formula. The theory of multidimensional residues is used as the main tool.\medskip

\noindent{\bf Keywords:} Grothendieck residue, Interpolation, Local algebra.
 \end{quote}
}


\section*{Introduction}
\label{s.Int}


The Lagrange and Hermite interpolation formulas are based on the interpretation of interpolation nodes as roots of suitable polynomials. Therefore, such formulas belong to the class of algebraic interpolations. The ideology of algebraic interpolations has recently received close attention from the point of view of interpolation theory for functions in several variables (see, for example \cite{Alpay}-\cite{YV}).

The article deals with the interpolation of functions in a complex space for which the sets of nodes are 0-dimensional analytic sets. In the one-dimensional case, these are Hermite interpolations. As the main apparatus, we use the theory of multidimensional residues.

Basic classical (standard) interpolations are understood as Lagrange, Hermite, Newton, etc. interpolations. Let's consider the first two of them.

\begin{problem*}[Lagrange]
Given a set of distinct points $\{w_j\}_{j=1}^m \subset \mathbb{C}$ and the values $c_j \in \mathbb{C}$, find the polynomial $f(z)$ of degree $m-1$ with the property $$f(w_j)=c_j,\quad j = 1, \ldots, m.$$
\end{problem*}

Note that the interpolation polynomial $f$ is defined in terms of the polynomial $p(z)=(z-w_1)\cdot\ldots\cdot(z-w_m)$ by the formula: $$f(z)=p(z)\sum\limits_{j=1}^m\frac{c_j}{z-w_j}\res\limits_{w_j}\left(\frac{1}{p}\right).$$
Thus, specifying the interpolation nodes in the form of the null set of the polynomial $p$ provides tools for constructing an interpolation polynomial by using residues. More general is the following

\begin{problem*}[Hermite]
Let $\{w_j\}_{j=1}^m\subset\mathbb{C}$ be a set of pairwise distinct points and the following values are given $$c_{j, \ell} \in \mathbb{C}, \quad\text{where } j = 1, \ldots, m,\quad\ell = 0, \ldots, \mu_j\,-\,1.$$ It is necessary to find a polynomial $f(z)$ of minimal degree, which at points $w_j$ has the given values of derivatives up to orders of $\mu_j-1$ inclusive, that is
\begin{equation}\label{asd}f^{(\ell)}(w_j)=c_{j, \ell}, \quad j = 1, \ldots, m, \quad \ell = 0, \ldots, \mu_j-1.\end{equation}
\end{problem*}

In the Hermite interpolation problem, it is advisable to enumerate the set of points $w_j$ taking into account their multiplicities, thereby considering the set $\{w_j\}_{j=1}^m$ as an algebraic set $p^{-1}(0)$, where \begin{equation}\label{polynomial} p(z) = (z-w_1)^{\mu_1}\cdot\ldots\cdot(z-w_m)^{\mu_m}.\end{equation}
The Hermite interpolation polynomial can be represented as

\begin{equation*}
\sum\limits_{j=1}^m  \frac{p(z)}{ (z-w_j)^{\mu_j}} \sum\limits_{\ell=0}^{\mu_j-1}\frac{c_{j, \ell}}{\ell !}\left(\sum\limits_{s=0}^{\mu_j-\ell-1}  (z-w_j)^{\ell+s} \res\limits_{w_j}\left(\frac{(z-w_j)^{\mu_j-1-s}}{p}\right)   \right),
\end{equation*}
that is, again with the participation of residues.

Our goal is to construct a similar variant for multidimensional interpolation. First, we will explain the main points of constructing an interpolation polynomial in the case of Lagrange interpolation. Let $\boldsymbol{p} = (p_1, \ldots, p_n)$ be a system of $n$ polynomials of $n$ variables $z= (z_1, \ldots, z_n)$ with a finite set of roots $\boldsymbol{p}^{-1}(0)\subset\mathbb{C}^n$. Assume that all roots are simple, this means that the Jacobian $J_{\boldsymbol{p}}$ of the system (mapping) $\boldsymbol{p}$ is not equal to zero at the roots:
$$J_{\boldsymbol{p}}(w)\neq 0, \quad \forall w\in \boldsymbol{p}^{-1}(0).$$


\begin{problem*}[Multidimensional Lagrange Interpolation]
For the given values $\{c_{w}\},$
$w\in\boldsymbol{p}^{-1}(0)$, find a polynomial $f(z)$ with the condition
\begin{equation}\label{one}
f(w)=c_{w}, \qquad w\in \boldsymbol{p}^{-1}(0).
\end{equation}
\end{problem*}

In the one-dimensional case, we can rewrite the Lagrange polynomial in the following form:
$$f(z) = \sum\limits_{w\in p^{-1}(0)} c_{w}\frac{p(z)}{z-w}\res\limits_{w}\left(\frac{1}{p}\right).$$
Therefore, we need to somehow interpret the division of the ideal $\langle p\rangle$, generated by the polynomial $p$ in the ring $\mathbb{C}[z]$, by its primary component $\langle z-w\rangle$ at the root of $w$ (about the primary components of ideal see \cite{VDV}). The polynomial $H(z, w)$ from the following decomposition is taken as such an interpretation
$$p(z)-p(\zeta) = (z - \zeta)H(z, \zeta).$$
So the Lagrange formula takes the form:
$$f(z) = \sum\limits_{w\in p^{-1}(0)} c_{w} H(z, w) \res\limits_{w}\left(\frac{1}{p}\right).$$

In the multidimensional case there is a system of identities
\begin{equation}
p_j(z)-p_j(\zeta) = \sum\limits_{k=1}^n (z_k-\zeta_k)h_{jk}(z, \zeta), \qquad j = 1,\ldots, n
\end{equation}
with a polynomial matrix $H(z, \zeta) = (h_{jk}(z,\zeta)).$
It is obvious that on the diagonal $\zeta = z$ the matrix $(h_{jk})$ coincides with the Jacobi matrix of the mapping $\boldsymbol{p}.$ At the roots $w\in\boldsymbol{p}^{-1}(0)$ the determinant $\det{H(z,w)}$ has the following properties:
\begin{itemize}
\item$\det{H(w,w)}$ is equal to the Jacobian $J_{\boldsymbol{p}}(w)$ of the mapping $\boldsymbol{p}$;
\item $\det{H(w, w')} = 0$ for each pair of different roots $w, w' \in \boldsymbol{p}^{-1}(0).$
\end{itemize}
The second property must be Lemma 4.1 in section 4.
Local residue (its exact definition is given in section 2) at the simple root $w\in\boldsymbol{p}^{-1}(0)$, that is at the root where $J_{\boldsymbol{p}}(w) \neq 0$, is calculated by the formula (\cite{Tsikh}, section 5.5):
$$\res\limits_{w} \left(\frac{h}{\boldsymbol{p}^I}\right) = \frac{h(w)}{J_{\boldsymbol{p}}(w)},$$
where $I = (1,\ldots, 1)$, and $\boldsymbol{p}^I = p_1\cdot\ldots\cdot p_n$

From these properties we get the following result:

\begin{theorem}
The polynomial
$$f(z) = \sum_{w \in \boldsymbol{p}^{-1}(0)} c_{w} \det{H(z, w)} \res\limits_{w}\left(\frac{1}{\boldsymbol{p}^I}\right),$$
where $\res\limits_{w}\left(\frac{1}{\boldsymbol{p}^I}\right)$ is a local residue, solves the problem (\ref{one})
\end{theorem}





\section{Multidimensional variant of Hermite interpolation}
\label{s.1}

Let's consider a multidimensional analogue of the Hermite problem.\linebreak  Let $\boldsymbol{p} = (p_1, \ldots, p_n)$ be a sequence of $n$ polynomials of $n$ variables $z= (z_1, \ldots, z_n)$ with a finite number of common roots $\boldsymbol{p}^{-1}(0)$. Obviously, $\boldsymbol{p}$ can be interpreted as mapping $\boldsymbol{p}: \mathbb{C}^n\to \mathbb{C}^n$. At the same time, $\boldsymbol{p}$ is associated with the ideals $\langle\boldsymbol{p}\rangle$ and $\langle \boldsymbol{p} \rangle_w$ generated by the sequence $p_j$ both in the ring $\mathbb{C}[z_1, \ldots, z_n],$ and in the local rings $\mathcal{O}_w$ of germs of holomorphic functions at the roots $w\in\boldsymbol{p}^{-1}(0)$. The quotient ring $\mathcal{O}_{w}/\langle\boldsymbol{p}\rangle_w$ has the structure of a vector space and is called a local algebra at the point $w$. Denote by $B_w$ the monomial basis of the local algebra $\mathcal{O}_{w}/\langle\boldsymbol{p}\rangle_w$ at the point $w\in \boldsymbol{p}^{-1}(0)$. The dimension of this algebra coincides with the multiplicity of the mapping $\boldsymbol{p}$ at the root of $w$ (see \cite{Tsikh}, section 19.4). The set of exponents $\ell$ in basic monomials $(z-w)^\ell\in B_w$ we denote as $A_w$.


\begin{problem*}[Multidimensional Hermite interpolation]
For the given values $\{c_{w, \ell}\}$, $w\in\boldsymbol{p}^{-1}(0); \ell \in A_{w}$, find a polynomial $f(z)$ with the condition
$$\frac{\partial^{|\ell|} f}{\partial z^\ell}(w)=c_{w, \ell}, \qquad w\in \boldsymbol{p}^{-1}(0), \qquad \ell \in A_{w},$$
where $\ell = (\ell_1, \ldots, \ell_n)$  and  $|\ell| = \ell_1+\ldots+\ell_n$.
\end{problem*}

Here we will consider a multidimensional version of the above problem in the case when the Newton polyhedra of the supports $A_{w}$ are essentially parallelepipeds $$\{0\leq\ell_1\leq d_1-1\}\times\ldots\times \{0\leq\ell_n\leq d_n-1\}.$$


To formulate the main theorem, we need the following definition of the\linebreak Grothendieck residue.

\begin{definition}{(see \cite{GH} or \cite{Tsikh})}
Let $h, p_1, \ldots,p_n$ be the holomorphic functions in the neighborhood $U_a$ of the point $a\in\mathbb{C}^n$, such that the mapping $\boldsymbol{p}=(p_1,\ldots,p_n)$ has an isolated zero at the point $a$: $\boldsymbol{p}^{-1}(0)\cap U_a = \{a\}$. By local residue or by Grothendieck residue of the meromorphic form $\omega=h\,dz/(p_1\cdots p_n)$ at the point $a$ we will call the integral $$\res\limits_{a}\omega=(2\pi i)^{-n}\int\limits_{\Gamma_a}\frac{h(z)\,dz_1\wedge\ldots\wedge dz_n}{p_1(z)\cdots p_n(z)},$$
over the cycle $$\Gamma_a=\{z\in U_a: |p_j(z)|=\varepsilon, j = 1, \ldots, n\},$$ where $\varepsilon$ is quite small, such that $\Gamma_a \Subset U_a;$ the orientation of the cycle $\Gamma_a$ is determined by the condition $$d(\arg p_1)\wedge\ldots\wedge d(\arg p_n)\geq 0.$$
\end{definition}

For the local residue, we will also use the notation $$\res\limits_a\left(\frac{h}{\boldsymbol{p}^I}\right),$$
assuming that $I = (1, \ldots, 1) \in \mathbb{Z}^n_+,$ and $\boldsymbol{p}^I = p_1\ldots p_n$. The main properties of local residues and methods of their calculations are given in \cite{GH}, \cite{Tsikh}, \cite{TY} and \cite{Tajima}.



Note that in the one-dimensional Hermite problem, in each root $w = w_j$ of a polynomial (\ref{polynomial}), it is required to restore $\mu_j$ values of derivatives (\ref{asd}), where $\mu_j$ is the multiplicity of the root $w_j$. In a multidimensional problem, the situation is the same. Recall (see \cite{Tsikh} or \cite{Milnor}) that the multiplicity of the isolated zero $w\in\mathbb{C}^n$ of the germ of the holomorphic map $\boldsymbol{p}:(\mathbb{C}^n, w)\to(\mathbb{C}^n, 0)$ is defined as the limit $$\mu_w = \overline{\lim\limits_{\xi\to w}} \# \{U_w \cap \boldsymbol{p}^{-1}(\xi)\},$$
where $\#$ is the power sign (cardinal) of the set, and $U_w$ is the neighborhood of a point, in the closure of which there are no roots of $\boldsymbol{p}$ other than $w$. The multiplicity $\mu_w$ coincides with the degree of mapping $z\to \boldsymbol{p}(z) / ||\boldsymbol{p}(z)||$ from the boundary $\partial U_w$ to the unit sphere (see \cite{Milnor}, Lemma B2).


In the one-dimensional case $(n=1)$, there are two equivalent definitions of multiplicity:
\begin{enumerate}
\item $z=w$ is a zero of multiplicity $d$ of the function $p(z)$ if function could be represented as
$$p(z) = (z-w)^{d} \varphi(z), \qquad \varphi(w) \neq 0;$$
\item $z=w$ is a zero of multiplicity $d$ of the function $p(z)$ if at the point $w$ the function $p$ and its derivatives up to the order of $d$ have the property
$$p(w) = p'(w) = \ldots = p^{(d-1)}(w) = 0, p^d(w) \neq 0.$$
\end{enumerate}

In a multidimensional situation $(n\geq 2)$ the concepts of multiplicity and order of zero differ: the first of them applies to the maps $\mathbb{C}^n\to \mathbb{C}^n$, and the second to the maps $\mathbb{C}^n\to \mathbb{C}^1$ (i.e. to functions). The concept of multiplicity is quite complex, there is no universal formula for its calculation. In order to solve the Hermite problem constructively (in the form of an explicit formula) we will consider the class introduced in the article \cite{LeTs} of polynomial maps $\boldsymbol{p}: \mathbb{C}^n\to\mathbb{C}^n$ which in each root $w\in\boldsymbol{p}^{-1}(0)$ have properties similar to the one-dimensional definition (ii). To describe these properties, the inequality $a\leq b$ of vectors $a, b\in\mathbb{Z}^n$ will be understood as the system of inequalities $a_i\leq b_i$ for coordinates of these vectors. Denote by $I$ the vector $(1, \ldots, 1) \in\mathbb{Z}^n$. The order of zero $w\in\boldsymbol{p}^{-1}(0)$ is the vector $d_w \in\mathbb{Z}_{+}^n$ such that

\begin{equation}\label{first}
\frac{\partial^{|\ell|} p_i}{\partial z^\ell}(w) = 0, \  0\leq \ell \leq d_w -I,
\end{equation}

\begin{equation}\label{second}
\det\left|\left|\frac{\partial^{d_{k}} p_i}{\partial z_k^{d_{k}}}(w)\right|\right| \neq 0\quad (\text{ here } i, k = 1,\ldots, n),
\end{equation}
where $d_k = {d_w}_k$ are the coordinates of the vector $d_w$.





\begin{theorem}\label{mainth}
Suppose that for each root $w\in\boldsymbol{p}^{-1}(0)$ there is such a vector  \linebreak $d_w \in\mathbb{Z}_{+}^n$, that the properties (\ref{first}) and (\ref{second}) are being fulfilled.
Then the polynomial $f(z)$ defined by expression
\begin{equation}\label{int}
\sum_{w \in \boldsymbol{p}^{-1}(0)} \det {H_w(z)} \left[ \sum_{\substack{\ell \leq d_w-I \\ k \leq d_w-I-\ell} } \frac{c_{w, \ell}}{\ell!}  (z-w)^{\ell+k}\res\limits_{w}\left(\frac{(z-w)^{d_w -I-k}}{\boldsymbol{p}^I}\right) \right]
\end{equation}
solves the multidimensional Hermite problem with nodes $w\in\boldsymbol{p}^{-1}(0)$, where \linebreak $H_w(z)~=~||h_{ik}||$ is a matrix from the representation
\begin{equation}\label{decomp}
\begin{pmatrix}
p_1(z)\\
\vdots\\
p_n(z)
\end{pmatrix}=
||h_{ik}||
\begin{pmatrix}
(z_1-w_{1})^{d_{1}}\\
\vdots\\
(z_n-w_{n})^{d_{n}}
\end{pmatrix}.
\end{equation}
\end{theorem}

Note that the existence of the representation (\ref{decomp}) will be proved by the lemma \ref{lem} below. We also note that when $w$ in (\ref{int}) is fixed, then the set $\{0\leq\ell\leq d_w-I\}$ of values of $\ell$ in (\ref{int}) means that $\ell$ runs over the parrallelepiped $A_w$ mentioned in statement of Hermite problem. So the cardinal $\# A_w = d_1 \cdot\ldots\cdot d_n$ and it is equal to the multiplisity $\mu_w = \mu_w(p)$ of the root $w$ when $p$ satisfies properties (\ref{first}) and (\ref{second}) (see \cite{LeTs}).


Consider the following example in $\mathbb{C}^2$, which satisfies the conditions of the theorem:
\begin{equation*}
\begin{cases}
p_1(z) = (z_1-1)^2+(z_2-1)^2+\frac{1}{3}(z_1-1)^2 (z_2-1)^2 \\
p_2(z) = (z_1-1)^2-(z_2-1)^2+\frac{2}{3}(z_1-1)^2 (z_2-1)^2.
\end{cases}
\end{equation*}
This mapping has five different roots: one of them $w_0 = (1, 1)$ of multiplicity 4 and four simple roots $w_j = (1\pm\sqrt{6}, 1\pm i\sqrt{2}), j = 1, 2, 3, 4$. The monomial basis of a local algebra at the root $w_0$ consists of four monomials: $1, z_1-1, z_2-1, (z_1-1) (z_2-1)$. Therefore, we need to find a polynomial $f(z)$ such that:
$$f(w_0) = c_{0,(0,0)}, f'_{z_1}(w_0) = c_{0,(1,0)}, f'_{z_2}(w_0) = c_{0,(0,1)}, f''_{z_1 z_2}(w_0) = c_{0,(1,1)},$$
$$f(w_1) = c_1, f(w_2) = c_2, f(w_3) = c_3, f(w_4) = c_4.$$

Let's find the matrix $H_{w_0}(z)$ from the condition:

\begin{multline*}
\begin{pmatrix}
p_1(z) \\
p_2(z)
\end{pmatrix} = 
\begin{pmatrix}
(z_1-1)^2+(z_2-1)^2+\frac{1}{3}(z_1-1)^2 (z_2-1)^2 \\
(z_1-1)^2-(z_2-1)^2+\frac{2}{3}(z_1-1)^2 (z_2-1)^2
\end{pmatrix}=\\
\begin{pmatrix}
(z_1-1)^2\left(1+\frac{1}{3}(z_2-1)^2\right)+(z_2-1)^2 \\
(z_1-1)^2\left(1+\frac{2}{3}(z_2-1)^2\right)-(z_2-1)^2
\end{pmatrix} = \\
\begin{pmatrix}
1+\frac{1}{3}(z_2-1)^2, &1\\
1+\frac{2}{3}(z_2-1)^2, &-1
\end{pmatrix}
\begin{pmatrix}
(z_1-1)^2 \\
(z_2-1)^2
\end{pmatrix} = H_{w_0}(z)
\begin{pmatrix}
(z_1-1)^2 \\
(z_2-1)^2
\end{pmatrix}.
\end{multline*}
Thus $\det{H_{w_0}(z)} = -2-(z_2-1)^2$. 
For $h = \sum\limits_{|\alpha|\geq 0} h_\alpha(z-w_0)^\alpha\in \mathcal{O}_{w_0}$, by the transformation formula of the local residue for mapping from $\boldsymbol{p}$ to $(H_{w_0}(z))^{-1}\boldsymbol{p}$, we~get (see \cite{Tsikh}, section 5.5)
$${\res\limits_{w_0}}\left(\frac{h}{p_1 p_2}\right) = {\res\limits_{w_0}}\left(\frac{h/ \det H_{w_0}(z)}{(z_1-1)^2(z_2-1)^2}\right) = -\frac{1}{2}h_{11}.$$
Since $d_{w_0} = (2, 2)$, we get that in (\ref{int}) the residues at $w_0$ are different from zero only when $k = (0, 0)$. Thus, the total contribution at the root $w=w_0$ in the formula (\ref{int}) is represented by a polynomial of the fourth degree
\begin{equation*}
f_0(z) = \left(-2-(z_2-1)^2\right)\left(\frac{-1}{2}\right)\sum\limits_{\ell \in \{0, 1\}^2}c_{0, \ell}(z_1-1)^{\ell_1}(z_2-1)^{\ell_2}
\end{equation*}

For simple zeros $w_j\in \boldsymbol{p}^{-1}(0)$ vector $d_{w_j} = (1, 1),$ therefore, the contributions they define to (\ref{int}) are given by one term for $\ell =k = (0, 0):$
$$f_j(z) = \det{H_{w_j}(z)}\cdot c_j\cdot\res\limits_{w_j}\left(\frac{1}{\boldsymbol{p}^I}\right)=c_j\frac{\det{H_{w_j}(z)}}{J_{\boldsymbol{p}}(w_j)}, \quad j = 1,2,3,4,$$
where $J_{\boldsymbol{p}}(w_j)$ is a value of the Jacobian of the mapping $\boldsymbol{p}$ at the point $w_j$. \linebreak For example, at the root $w_1 = (1+\sqrt{6}, 1+i\sqrt{2})$ determinant $\det{H_{w_1}(z)}$ \linebreak is equal to
\begin{multline*}
- z_{1} z_{2}^{3} + z_{1} z_{2}^{2}(3- \sqrt{2} i)+z_{1} z_{2}(- 3 + 2 \sqrt{2} i) + z_{1}(1 - \sqrt{2} i) +z_{2}^{3}(1- \sqrt{6}) +\\
z_{2}^{2}(3 \sqrt{6}-3 - 2 \sqrt{3} i+\sqrt{2} i) + z_{2}(3- 3 \sqrt{6} - 2 \sqrt{2} i+ 4 \sqrt{3} i) +(- 1 + \sqrt{6} - 2 \sqrt{3} i + \sqrt{2} i)
\end{multline*}

Calculations show that
$$\det{H_{w_1}(w_1)} = J_{p}(w_1) = 16 i\sqrt{3},$$
so contribution $f_1(z) = -i c_1[\det{H_{w_1}(z)}]/16\sqrt{3}.$



\section{Structure of local ideals $\langle\boldsymbol{p}\rangle_w$ under conditions (\ref{first}) and (\ref{second})}


Recall that the mapping $\boldsymbol{p} = (p_1, \ldots, p_n)$ at each point $w\in\boldsymbol{p}^{-1}(0)$ defines the ideal $\langle\boldsymbol{p}\rangle_w$ of the ring of germs $\mathcal{O}_w$ generated by $p_1, \ldots, p_n$. For brevity, we will assume that $w=0$.
\begin{lemma}\label{lem}
If the conditions (\ref{first}), (\ref{second}) are met for the germs $p_1,\ldots, p_n\in \mathcal{O}_0$,
%\begin{equation}\label{eq1}
%\frac{\partial^{\,{\boldsymbol{\alpha}}} Q_i}{\partial z^{\boldsymbol{\alpha}}}(0) = 0, \qquad 0 \leq \alpha_i \leq d_i - 1, \qquad i = 1,\ldots,n,
%\end{equation}
%
%\begin{equation}\label{eq2}
%\det\left[
%\frac{\partial^{\,d_j} Q_i}{\partial z_j^{d_j}}(0)
%\right] \neq 0.
%\end{equation}
then in the ring $\mathcal{O}_0$ there is equality of ideals:
\begin{equation}
\langle p_1, \ldots, p_n \rangle = \langle z_1^{d_1}, \ldots, z_n^{d_n} \rangle.
\end{equation}

\end{lemma}
\begin{proof}
Consider in $\mathbb{R}^n$ a simplex $S$ stretched over its vertices $v_j = (0, \ldots, d_j, \ldots, 0)$, $j =1,\ldots, n.$
From the condition (\ref{first}) it follows that the slices $p_i$ on $S$ constitute a set of initial weighted homogeneous polynomials for $p_i$:
$$(p_i)_{*} = m_{i 1}z_1^{d_1} + \ldots + m_{i n}z_n^{d_n}, \qquad i = 1, \ldots, n,$$
with weight $\delta = (\delta_1,\ldots, \delta_n)$, where $\delta_j = 1/d_j$. 
By condition (\ref{second}), the matrix $M = (m_{i j})$ is non-degenerate. By converting $M^{-1}$, the system $p_1, \ldots, p_n$ is reduced to the form $$P_i =z_i^{d_i}+g_i(z), \quad i =1, \ldots, n,$$
where each monomial $z^\beta$ of the germ $g_i$ has $\delta$--weighted degree of $\langle\beta, \delta\rangle$ is greater than that of the monome $z_i^{d_i}$. In this case , the equalities are fulfilled
\begin{equation}\label{34}
\frac{\partial^{\boldsymbol{\alpha}} P_i}{\partial z^{\boldsymbol{\alpha}}}(0) = 0
\text{\ \ for all \ \ } \boldsymbol{0}\leq \boldsymbol{\alpha} \leq \boldsymbol{d}-\boldsymbol{I}.
\end{equation}
Due to the non-degeneracy of $M$, there is equality of ideals
\begin{equation*}
\langle Q_1, \ldots, Q_n \rangle = \langle P_1, \ldots, P_n \rangle.
\end{equation*}
And from now on we will prove the equality of ideals
\begin{equation}\label{star}
\langle P_1, \ldots, P_n \rangle = \langle z_1^{d_1}, \ldots, z_n^{d_n} \rangle.
\end{equation}

 

It follows from the equalities (\ref{34}) that any monomial of $P_i$ is divisible by at least one element from the set $\{z_1^{d_1}, \ldots, z_n^{d_n}\}.$ Thus,
$$P_i= z_1^{d_1}\cdot f_{i 1}+z_2^{d_2}\cdot f_{i 2} +\ldots +z_n^{d_n}\cdot f_{i_n}.$$
Therefore, there is direct inclusion of ideals in (\ref{star}). 

Let's prove the inverse inclusion, that is, that each $z_i^{d_i}$ belongs to the left part of (\ref{star}). To do this, we will use the local duality theorem from the theory of multidimensional residues (see \cite{Tsikh}, section 5.5). According to this theorem, $z_i^{d_i}\in \langle P_1, \ldots, P_n\rangle$ if and only if
$${\res\limits_0}\left(\frac{z_i^{d_i} h}{\boldsymbol{P}^I}\right) = 0, \qquad \forall h \in \mathcal{O}_0.$$
Without limiting generality, we will conduct a proof for $i=1$ and first consider the case when $h = 1$:
\begin{equation}\label{resid}
{\res\limits_0}\left(\frac{z_1^{d_1}}{\boldsymbol{P}^I}\right) = \frac{1}{(2\pi i)^n}\int\limits_\Gamma \frac{z_1^{d_1} dz}{P_1(z)\cdots P_n(z)},
\end{equation}
where $\Gamma = \Gamma_0 = \{z\in U_0: |P_j(z)| = \varepsilon_j, j = 1,\ldots,n \}.$ Recall that $U_0$ is a neighborhood of $0$ containing no other roots of $\boldsymbol{P}$ in its closure.

Note that outside the set of complex hypersurfaces \\
$\{P_1(z) = 0\}, \ldots, \{P_n(z) = 0\}$ the cycle $\Gamma_0$ is homologous to the torus
$$\gamma_0 = \{|z_1| = r^{\delta_1}, \ldots, |z_n| = r^{\delta_n} \}, \qquad r << 1.$$
In fact, since the monomials of germs $g_i$ have $\delta$--weighted degree greater than $z_i^{d_i}$, then $$\left.|z_i^{d_i}|\right|_{\gamma_0} =r>o_i(r) = \left.|g_i(z)|\right|_{\gamma_0},$$
where $o_i(r)$ are infinitesimal quantities relative to $r$. Therefore $P_i = z_i^{d_i}+ g_i \neq 0$ on $\gamma_0$, and according to (\cite{YA}, Lemma 4.9) for small $r$, the cycle $\Gamma_0$ is homologous to the cycle $\gamma_0$ in $U_0\backslash \{P_1 \cdot\ldots\cdot P_n = 0\}.$ Returning to the local residue (\ref{resid}), we will study it as an integral over the cycle $\gamma_0$:

\begin{equation*}
\begin{split}
 &\frac{1}{(2\pi i)^n}\int\limits_{\gamma_0} \frac{z_1^{d_1} dz}{(z_1^{d_1} + g_1(z))\cdots (z_n^{d_n} + g_n(z))} =\\
&= \frac{1}{(2\pi i)^n}\int\limits_{\gamma_0} \frac{z_1^{d_1} dz}{z_1^{d_1}\cdots z_n^{d_n}(1 + \frac{g_1(z)}{z_1^{d_1}})\cdots (1+ \frac{g_n(z)}{z_n^{d_n}})} = \\
&=\frac{1}{(2\pi i)^n}\int\limits_{\gamma_0} \sum\limits_{\boldsymbol{k}\geq \boldsymbol{0}}\frac{\left(-\frac{g_1(z)}{z_1^{d_1}}\right)^{k_1}\cdots \left(-\frac{g_n(z)}{z_n^{d_n}}\right)^{k_n} dz}{z_2^{d_2}\cdots z_n^{d_n}} = \\
&=\frac{1}{(2\pi i)^n} \sum\limits_{\boldsymbol{k}\geq \boldsymbol{0}} \int\limits_{\gamma_0} \frac{(-1)^{|\boldsymbol{k}|}g_1^{k_1}\cdots g_n^{k_n} dz}{z_1^{d_1 k_1} z_2^{d_2(k_2+1)}\cdots z_n^{d_n(k_n+1	)}}.
\end{split}
\end{equation*}


Here, the second equality is performed based on the geometric progression formula.
The last integrals give a nonzero contribution only for such monomials $z^\alpha$ in the numerator, for which
$$\alpha = (d_1 k_1 -1, d_2(k_2+1) -1, \ldots, d_n(k_n+1) -1).$$
A monomial from the multiplier $g_i^{k_i}$ can be obtained only from the product of $k_i$ monomials of the germ $g_i$. Therefore, we have $|k| := k_1+k_2+\ldots+k_n$ monoms of our germs, from which the numerator is obtained. But each monomial of the germ, as we showed above (based on the property (\ref{34})), is divided by at least one element from the set $\{z_1^{d_1}, \ldots, z_n^{d_n}\}.$ The monomials that are divisible by $z_1^{d_1}$ must be less than $k_1$, because otherwise in the numerator the power of $z_1$ will be greater than or equal to $d_1 k_1$, but we should get $d_1 k_1 -1$. Therefore, monomials that are divided into some element of the set $\{z_2^{d_2}, z_3^{d_3}, \ldots, z_n^{d_n}\}$, we have at least $k_2+k_3+\ldots+k_n+1$. Now suppose that for any $i \in \{2, \ldots, n\}$ the number of monomials divisible by $z_i^{d_i}$ is not more than $k_i$. In this case, the monomials that are divided into some element of the set $\{z_2^{d_2}, z_3^{d_3}, \ldots, z_n^{d_n}\}$, we have no more than $k_2+k_3+\ldots+k_n$. We get a contradiction, therefore there exists such $i$ that the monomials that are divisible by $z_i^{d_i}$ are greater than $k_i$. But in this case, in the numerator we will get a power of $z_i$ not less than $d_i(k_i+1)$. And this is more than the required degree $d_i(k_i+1)-1$. Thus, in any case, for some variable, the final degree will be greater than $-1$, and this will lead to the zeroing of the integral (\ref{resid}) corresponding to $h=1$.

Multiplying the numerator $z_1^{d_1}$ in the integral (\ref{resid}) by some irreversible germ $h$ (i.e. $h(0)=0$) will increase its order of zero at the point $z= 0$, so $${\res\limits_0}\left(\frac{z_1^{d_1}h}{\boldsymbol{P}^I}\right) = 0, \qquad \forall h \in\mathcal{O}_0.$$
This is exactly what needed to be shown.

 


\end{proof}



\section{Proof of the theorem \ref{mainth}}

To prove the main theorem, we will need, along with the lemma \ref{lem}, the following statement.


\begin{lemma}({\cite{Tsikh}, section 5.5}) \label{germs}
Let $\boldsymbol{q}, \boldsymbol{g}: G\to\mathbb{C}^n$ be two holomorphic maps of the domain $G \subset \mathbb{C}^n$ associated by the relation $\boldsymbol{g}= A\boldsymbol{q}$ with the functional matrix $A =||a_{ij}(z)||$, $a_{ij}\in\mathcal{O}(G).$ If for some point $w\in G$
$$\boldsymbol{q}(w)\neq 0, \boldsymbol{g}(w) = 0,$$ then $\det A$ belongs to the ideal $I_w(\boldsymbol{g})\subset\mathcal{O}_w$ generated in the ring of germs of holomorphic functions by the system $\boldsymbol{g}=(g_1, \ldots, g_n).$
\end{lemma}

\begin{proof}[Proof of the theorem \ref{mainth}]

If in the lemma \ref{germs} we put $\boldsymbol{g} = \boldsymbol{p}, \boldsymbol{q} =(z-w)^{d_{w}}$, then we get that for $w\in \boldsymbol{p}^{-1}(0)$ the determinant $ \det{H_w(z)}$ belongs to the local ideals $I_{w'}(\boldsymbol{p}), w'\in \boldsymbol{p}^{-1}(0)\backslash\{w\}$. Therefore, these ideals will also have an whole term equal to the product of $\det{H_w(z)}$ by the square bracket in (\ref{int}). Thus, this term will be vanished at points $w'$ under the action of the corresponding differential operators $\left\{\frac{\partial^{|\ell|}}{\partial z^\ell}\right\}_{\ell\in A_{w}}$ from the condition $\ref{first}$. In other words, this term will not give a contribution for the remaining points.

Now we show that it gives the necessary contribution at the point $z=w$. To do this, first learn how to calculate local residues $\res\limits_w\left(1/\boldsymbol{p}^I\right)$ relative to our mapping $\boldsymbol{p}$. The following local residue transformation formula will help us (see \cite{Tsikh}, section 5.5) when we are moving from the mapping $\boldsymbol{q}$ to the mapping $\boldsymbol{g}= A\boldsymbol{q}$: if $w$ is an isolated zero of the mappings $\boldsymbol{q}$ and $\boldsymbol{g}$, then for any germ $\varphi \in\mathcal{O}_w$

$${\res\limits_w}\left(\frac{\varphi}{\boldsymbol{q}^I}\right) = {\res\limits_w}\left(\frac{\varphi \det A}{\boldsymbol{g}^I} \right).$$


In our case, based on the decomposition (\ref{decomp}), we have $$\boldsymbol{q} = \boldsymbol{p}, \boldsymbol{g}= ((z-w_1)^{d_{w_1}}, \ldots, (z-w_n)^{d_{w_n}}), A = {H_w^{-1}(z)}$$. Therefore, according to the transformation formula , we get:
$${\res\limits_{w}}\left( \frac{\varphi}{\boldsymbol{p}^I} \right) = {\res\limits_{w}}\left(\frac{\varphi \det H_w^{-1}(z)}{\boldsymbol{g}^I}\right) = {\res\limits_{w}}\left(\frac{\varphi}{\boldsymbol{g}^I \det H_w(z)}\right).$$


Residue relative to $\boldsymbol{g}$ is easily calculated: for a germ $\varphi = \sum\limits_{|\alpha|\geq 0} \varphi_\alpha(z-w)^\alpha$ it is equal to the coefficient $\varphi_\alpha$ with the index $\alpha = {(d_{w}-I)}$. Therefore, it is enough for us to decompose the germ $\frac{1}{\det H_w(z)}$ into a Taylor series centered at $z = w$:
$$\frac{1}{\det H_w(z)} = \sum\limits_{|\alpha|\geq0} h_{w,\alpha} (z_1-w_{1})^{\alpha_1}\ldots (z_n-w_{n})^{\alpha_n}.$$

Next, we show that for a fixed $\ell$ the multiplier for $\frac{c_{w,\ell}}{\ell!}(z_1-w_{1})^{\ell_1}\cdot (z_n-w_{n})^{\ell_n}$ in (\ref{int}) has the form
\begin{equation}\label{vid}
1+\text{ an element from the ideal } \langle\bold{(z-w)^{d_w-\boldsymbol{\ell}}}\rangle,
\end{equation}
where $\langle\bold{(z-w)^{d_w-\boldsymbol{\ell}}}\rangle$ is an ideal generated by the system $$(z_1-w_1)^{d_1-\ell_1}, \ldots, (z_n-w_n)^{d_n-\ell_n}.$$ This will just mean that this term will only give a contribution to the derivative of~a fixed order $\ell$, all other derivatives will vanish due to the residual term. Indeed, since $\boldsymbol{g}^I = (z_1-w_1)^{d_1}\ldots(z_n-w_n)^{d_n}$, according to the Cauchy formula, the residue in (\ref{int}) is $h_{w,k}$, so this multiplier is equal to

\begin{multline*}
 \det {H_w(z)} \cdot
\sum_{k \leq d_{w} -I-\ell} (z-w)^k \res_{w}\left( \frac{(z-w)^{d_{w} -I-k}}{\boldsymbol{g}^I \det{H_w(z)}}\right) =\\
 \det {H_w(z)} \cdot
  \sum_{k \leq d_{w} -I-\ell} h_{w, k} (z-w)^k  =  \\
 \det {H_w(z)} \cdot \left(\frac{1}{\det H_w(z)} + \text{ an element from the ideal } \langle\bold{(z-w)^{d_w-\boldsymbol{\ell}}}\rangle \right),
\end{multline*}
that is, it has the form (\ref{vid}).


When multiplying the remainder in (\ref{vid}) by our multiplier $$\frac{c_{w,\ell}}{\ell!}(z_1-w_{1})^{\ell_1}\cdots (z_n-w_{n})^{\ell_n}$$ we will get an element from the ideal $\langle\bold{(z-w)^{d_w-\boldsymbol{\ell}}}\rangle$. All derivatives of the orders ${\ell'} \in A_{w}\backslash \{\ell\}$ will be zeroed on this element at the point $w$.
The derivative of the order $\ell$ will be equal to $c_{w, \ell}$, which was required.

\end{proof}





\smallskip
{\bf Acknowlegement.} 
The investigation was supported by the Russian Science Foundation, grant No. 20-11-20117









\bigskip

\end{document}
